Author’s response to reviews

Title: Morbidity and health-related quality of life of patients accessing laparoscopic sleeve gastrectomy: a single-centre cross-sectional study in one province of Canada

Authors:

Laurie Twells (ltwells@mun.ca)

Shannon Driscoll (sdriscoll@mun.ca)

Deborah Gregory (dgregory@mun.ca)

Kendra Lester (klester@mun.ca)

John Fardy (jfardy@mun.ca)

Dave Pace (dpace@mun.ca)

Version: 2 Date: 27 Nov 2017

Author’s response to reviews:

Dear BMC Obesity Reviewers,

Thank you for reviewing our article and providing minor editorial changes. I have addressed all the comments below:

Reviewer comment:
1) Title page: please provide the email addresses of all authors

Author response: Done, this information has been added to the Title page.

Reviewer comment:
2) Declarations section:
   (i) please move this to appear after the Conclusions section and before the Reference list

Author response: Done, these statements have been moved to after the Conclusions section.

Reviewer comment:
(iii) Consent: please change this heading to Consent to publish. Please provide Not applicable under the heading.

Author response: Done and in the section of Declarations after the Conclusions section.

Reviewer comment:
(iv) Authors contributions: please provide the specific contribution of DP to the study.

Author response: The contribution of DP has been added to Author Contributions in the Declarations section after the Conclusions section. [please see Editorial Policies below for the correct heading names]

Editorial comments:
3) Abstract:
   (i) the Abstract should be provided in the main Word file

Author response: The abstract has been added to the word file.

(ii) Please change 'Interpretation' to Conclusions

Author response: Done in both the uploaded version and the abstract added to the word file.

4) Introduction should be changed to Background.

Author response: Done in the main paper.

We appreciate all the reviewer and editorial comments. yours sincerely,